

# **JGR** Biogeosciences

## **RESEARCH ARTICLE**

10.1029/2021JG006344

#### **Key Points:**

- We present old <sup>14</sup>C dates from waters draining eroded temperate peats, caused by deep C loss and low water tables in gullied peats
- Novel approaches to dissolved organic carbon source identification indicate that water table drawdown and the nature of peat erosion influence loss of old carbon
- Our method has potential as a rapid way to indicate peatland function and assess restoration success

#### **Correspondence to:**

M. G. Evans and D. M. Alderson, martin.g.evans@manchester.ac.uk; danielle.alderson@manchester.ac.uk

#### Citation:

Evans, M. G., Alderson, D. M., Evans, C. D., Stimson, A., Allott, T. E. H., Goulsbra, C., et al. (2022). Carbon loss pathways in degraded peatlands: New insights from radiocarbon measurements of peatland waters. *Journal of Geophysical Research: Biogeosciences*, *127*, e2021JG006344. https://doi. org/10.1029/2021JG006344

Received 29 MAR 2021 Accepted 13 JUN 2022

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## Carbon Loss Pathways in Degraded Peatlands: New Insights From Radiocarbon Measurements of Peatland Waters

Martin G. Evans<sup>1</sup>, Danielle M. Alderson<sup>1</sup>, Chris D. Evans<sup>2</sup>, Andrew Stimson<sup>3</sup>, Timothy E. H. Allott<sup>1</sup>, Claire Goulsbra<sup>1</sup>, Fred Worrall<sup>4</sup>, Tia Crouch<sup>5</sup>, Jonathan Walker<sup>6</sup>, Mark H. Garnett<sup>7</sup>, and James Rowson<sup>8</sup>

<sup>1</sup>Upland Environments Research Unit, Department of Geography, University of Manchester, Manchester, UK, <sup>2</sup>UK Centre for Ecology and Hydrology, Bangor, UK, <sup>3</sup>North Pennines Area of Outstanding National Beauty Partnership, Stanhope, UK, <sup>4</sup>Department of Earth Sciences, University of Durham, Durham, UK, <sup>5</sup>Moors for the Future Partnership, Edale, UK, <sup>6</sup>Swansea University, Swansea, UK, <sup>7</sup>NERC Radiocarbon Facility, East Kilbride, UK, <sup>8</sup>Edge Hill University, Ormskirk, UK

**Abstract** Peatland carbon stores are under widespread anthropogenic pressure, resulting in degradation and carbon loss. This paper presents  $DO^{14}C$  (Dissolved Organic Carbon) dates from waters draining two eroded blanket peatland catchments in the UK. Both catchments are characterized by severe gully erosion but one additionally has extensive surface erosion on unvegetated surfaces.  $DO^{14}C$  values ranged from 104.3 to 88.6 percent modern (present to 976 Before Present). The oldest DOC dates came from the catchment characterized by both gully and surface erosion and are among the oldest reported from waters draining temperate peatlands. Together with peat age-depth data from across the peatland landscape, the  $DO^{14}C$  ages identify where in the peat profile carbon loss is occurring. Source depths were compared with modeled water table data indicating that in the catchment where gully erosion alone dominated, mean water table was a key control on depth of DOC production. In the system exhibiting both gully erosion and surface erosion, DOC ages were younger than expected from the age of surficial peats and measured water tables. This may indicate either that the old organic matter exposed at the surface by erosion is less labile or that there are modifications of hydrological flow pathways. Our data indicate that eroded peatlands are losing carbon from depth, and that erosion form may be a control on carbon loss. Our approach uses point measurements of  $DO^{14}C$  to indicate DOC source depths and has the potential to act as an indicator of peatland function in degraded and restored systems.

**Plain Language Summary** Peatlands are the largest terrestrial soil carbon store, but are under widespread anthropogenic pressure, resulting in degradation and carbon loss. In this study of UK upland peatlands, we used radiocarbon dating of dissolved organic carbon (DOC) from waters draining eroded peatlands to identify the age of the carbon that has been lost. These dates are among the oldest reported from waters draining temperate peatlands. The dates combined with age-depth data from across the peatland landscape allowed us to model where in the peat profile the carbon loss was concentrated. We compared these results with water table data and found that in the systems we studied, the type of peatland erosion that the system had experienced was an important factor controlling the age of carbon lost. The close association between radiocarbon age of carbon and water table suggested that radiocarbon dating of DOC might be a useful catchment scale proxy for water table, a direct measurement of the locus of carbon loss from the peat, and an effective measure of the overall "health" of the peatland ecosystem. The approach therefore has potential as a rapid method of assessing stabilization and restoration of intact peatland function.

## 1. Introduction

Peatlands are the largest terrestrial soil carbon store, containing circa 620 Pg of carbon (Dargie et al., 2017; Page et al., 2011). Deep peat accumulations have been built up over at least Holocene timescales (Gorham et al., 2012; Harden et al., 1992), so that the long-term accumulation of organic matter represents an important component of the terrestrial carbon cycle. However, peatlands are under widespread pressure with stressors, such as plantation forestry and agriculture (Dommain et al., 2016), drainage (Pronger et al., 2014), nitrogen pollution (Payne, 2014), controlled and uncontrolled fires (Turetsky et al., 2015), overgrazing of seminatural areas, thawing of permafrost (Frey & Smith, 2005; Romanovsky et al., 2012; Schuur & Mack, 2018), and changing hydroclimate (Clark et al., 2010). This diverse set of issues has the potential to fundamentally alter the peatland carbon store with

2022, 7, Downloade

most anthropogenic pressures leading to carbon loss (Page & Baird, 2016) and accelerating climate warming (Gallego-Sala et al., 2018).

Peatlands that have been subject to these pressures often feature low water tables, resulting in enhanced losses of old carbon from peatland storage as fluvial carbon in dissolved form, in addition to direct losses to the atmosphere through in situ oxidation of organic carbon. The primary control on decomposition rates in peatlands and so on gaseous and dissolved organic carbon (DOC) production is the availability of oxygen to support aerobic microbial decomposition, which is in turn significantly influenced by water table (Clymo, 1984; Strack et al., 2008). Globally, direct gaseous carbon emissions are enhanced by peatland degradation (e.g., an additional 1.30 GtCO<sub>2</sub>/ yr released to the atmosphere due to drainage in 2008; Joosten, 2010 and 0.2 Gt CO<sub>2</sub> eq/yr released due to peatland fires; van der Werf et al., 2010). Fluvial dissolved carbon losses in both pristine and degraded systems occur through the mobilization of soluble products of organic matter decomposition in the aerobic upper layers of the peat (typically the top 10–20 cm in intact peats). In peatland soils, there is commonly a rapid (often exponential) decline in hydraulic conductivity with depth (Fraser et al., 2001; Hoag & Price, 1995; Morris et al., 2015) so that near-surface flow paths dominate runoff (Holden & Burt, 2003a). Limited evidence suggests that fluvial carbon losses are increased by drainage and/or other activities that lower the water table and increase the depth of the oxic zone (Evans et al., 2016; Moore et al., 2013). Lower water tables may also modify the near-surface hydrology leading to matrix throughflow and macropore flow in deeper peats (Holden et al., 2006).

In heavily eroded systems, therefore, losses of DOC may occur much deeper in the peat profile. Loss of carbon from the peatland surface is associated with the turnover of surface vegetation and fresh litter, whereas carbon loss from deeper peats due to peatland drainage represents an additive transfer of carbon from the long-term peatland store to the atmosphere.

Peatland soils are cumulative deposits so that the age of the organic remains in the peat profile is correlated with depth. Aging carbon in runoff from degraded peatland systems therefore offers the potential to fingerprint where in the peat profile decomposition is occurring. Bulk radiocarbon dates from stream water reflect the age of a mixture of the organic material that has been leached into the fluvial system. Aged signals can therefore be diluted by modern contributions (turnover of litter), but old dates will definitively indicate significant contributions from the decomposition of old organic matter stored in the peat profile. DOC ages can also be dependent on water residence times (Dean, 2019) although this is unlikely to be a significant issue in the blanket peat systems. In these peatlands, water residence times are often short (minutes) as they tend to be hydrologically "flashy" systems (Shuttleworth et al., 2019), generating the majority of flow from surficial flow paths so that DOC from slow flow paths is a minor component of stormflow. Older DOC radiocarbon ages from waters draining degraded peatlands are an indicator of modified carbon cycling, and potentially accelerated, loss of stored carbon (Evans, Bonn, et al., 2014).

Despite the compilation of large data sets of dated fluvial DOC (e.g., Butman et al., 2015; Marwick et al., 2015), data from peatlands are relatively limited, particularly from heavily degraded peatlands. Dating of DOC in waters leaving intact temperate peatlands has usually produced modern radiocarbon dates, consistent with the diplotelmic model of peat soils (Evans et al., 2007; Raymond et al., 2007) and turnover of modern organic matter at the peatland surface. However, at a few drained and degraded peatland sites, much older radiocarbon dates have been recorded. Evans, Page, et al. (2014) have reported radiocarbon as old as 4,374 years Before Present (BP) (59.4% modern) for drained tropical peats. Northern hemisphere drained peatlands produced aged carbon typically in the range 100–300 years BP with one sample from the UK Peak District dated as 2006 years BP (Evans, Page, et al., 2014). Hulatt et al. (2014) have reported radiocarbon ages of up to 3,100 years BP from agricultural peatlands and peatland extraction sites in Finland.

In this paper, we have assembled radiocarbon ages on DOC samples from what are arguably the most severely eroded blanket peatlands in the world; the Peak District in the UK (Evans & Warburton, 2007). These include radiocarbon data from published and unpublished projects (Evans et al., 2013; Stimson et al., 2017) and some new data. By comparing dates from two sites with contrasting erosional characteristics, and further comparing DOC ages to measured and modeled water tables in these catchments, we aim to understand the controls on the age of DOC generated and exported from eroded peatland catchments. Furthermore, we aim to assess the implications for the use of <sup>14</sup>C as an indicator of deep peat stability and as a marker for restoration recovery. We test the



## Journal of Geophysical Research: Biogeosciences



**Figure 1.** Hypotheses linking erosion and production of older dissolved organic carbon (DOC) from gullied peatlands. The blue line is the position of the water table and the brown arrow is the direction of the movement of DOC. The black line represents the gully profile and the surface of the peat. (a) Water table as the control: older DOC as water tables lowered; (b) supply of modern DOC from fresh litter and root exudates is the control: older DOC as this supply is removed; (c) presence of surficial erosion is the control: older DOC generated at the surface as younger peat is removed by erosion.

primary hypothesis that these degraded peatlands are losing carbon at depth. In addition, we will consider three hypotheses that could explain the production of old DOC from eroding peatland systems. These are (Figure 1):

- 1. That decomposition of older carbon is associated with lower water tables, which results from gully erosion of the peat (Figure 1a)
- 2. Extensive surface erosion can remove vegetation so that there is bare peat at the surface, which reduces supply of "modern" DOC from freshly degraded plant litter and root exudates erosion (Figure 1b)
- 3. Where there has been considerable depth of surface erosion, old peat at the surface might contribute old carbon even under relatively high water table conditions (Figure 1c).

## 2. Study Area and Methods

## 2.1. Study Site

In the UK, peatland carbon storage has been estimated at around 5 Gt C, which represents 52% of total UK soil carbon storage (Ostle et al., 2009). However, many UK peatlands are severely degraded with extensive gully erosion occurring in many parts of upland Britain (Evans & Warburton, 2007). The most severe erosion is seen in the southern Pennine region, which has been described as the "badlands of Britain" (Tallis, 1997). This description reflects their past history of relatively intensive land use, including grazing, drainage, and the widespread occurrence of prescribed burning and wildfires, as well as a history of acidifying atmospheric pollution resulting from their proximity to the major industrial cities of Northern England, which led to the loss of key peatland vegetation including Sphagnum mosses (Evans et al., 2006; Tallis, 1987). Previous dates on fluvial DOC have been reported in this region (Evans, Page, et al., 2014). Dates from an ungullied site and a re-wetted site were 99.12% and 101.43% modern, modeled as age equivalents of 399 and 291 years BP, respectively, while two dates from gullied sites were 95.11% and 81.14% modern, modeled as age equivalents of 655 and 2006 years BP (Evans, Page, et al., 2014).

This study focused on two degraded headwater peatland catchments in the Peak District National Park of the Southern Pennines. These blanket bogs are ombrotrophic mires that closely follow the underlying topography on flat surfaces or gentle slopes. The peats develop under specific climatic conditions, notably a minimum of 1,000 mm rainfall, mean temperatures of  $<15^{\circ}$ C, and little seasonal variability (Lindsay et al., 1988). Blanket peats have distinctive flora and fauna, and their ecological functioning is intimately linked with their hydrology (Charman, 2002; Evans, Bonn, et al., 2014).





**Figure 2.** Study sites location and sampling points. (a) Location of Bleaklow Plateau and Kinder Scout in relation to the UK and each other. (b) Hillshade digital elevation model identifying the two study catchments and green dots representing the location of the sampling points.

The study sites are within a blanket mire complex, which span two upland plateaux about 5 km apart and which form the watershed dividing east and west flowing drainage in the southern Pennines (Figure 2). The Kinder River catchment drains the northern half of the Kinder Scout Plateau and the Upper North Grain stream catchment drains the southern flanks of the Bleaklow Plateau (Figure 2). Both catchments are areas of upland blanket peat at elevations between ~500 and 633 m (Shuttleworth, 2015; Stimson, 2015). Upper North Grain descends ~60 m (Goulsbra et al., 2014) before a confluence with the River Ashop, which eventually drains into Ladybower Reservoir, whereas the Kinder River headwaters descend almost 300 m into Kinder Reservoir (Stimson, 2015). These are small upland streams with stream depths less than a meter in high flow and maximum stream widths on the order of five m.

The mean annual temperature in the area is  $6.9^{\circ}$ C with a range of -8 to  $22^{\circ}$ C, over the period 2004–2013, and the mean annual rainfall is ~1,300 mm (Clay & Evans, 2017). The vegetation is dominated by wavy-hair grass



b) Kinder River 16 14 10 12 Frequency 10 Frequency 6 8 6 4 2 00 0 50 00 18 20 th 30 Peat Depth cn

a) Upper North Grain

Peat Depth cm

Figure 3. Hillshade Digital Elevation Model of study catchments with the gully network highlighted. Photos of the nature of erosion and peat depth distributions. (a) Gully erosion in the Upper North Grain (UNG) catchment. (b) Extensive bare peat and sheet erosion in the Kinder River catchment.

> (Deschampsia flexuosa), cottongrass (Eriophorum vaginatum), heather (primarily Calluna vulgaris) bilberry (Vaccinium myrtillus), and crowberry (Empetrum nigrum). A number of native UK blanket bog forming species were historically depleted in the area due to its past pollution and management history, but key indicator species including bryophytes (e.g., peat-forming Sphagnum spp.) have seen improvements in coverage over the past 10 years as a consequence of restoration efforts in both areas (Alderson et al., 2019).

> Stream waters were sampled for radiocarbon analysis of DOC at three sites in the Upper North Grain catchment and two sites in the Kinder River catchment (Figure 2). In the Kinder River system, one sample was taken at the outlet of the peat catchment and one further sample from a short distance downstream circa 500 m. Sampling sites in the Upper North Grain catchment were all within the study catchment. This study utilized available radiocarbon data from other projects and published works (Evans et al., 2013; Stimson et al., 2017 and unpublished data) and these studies suggest that there was no systematic downstream change in observed <sup>14</sup>C ages over short distances such as these.

> Figure 3 shows each catchment in more detail, highlighting the differences in the nature of the erosion and peat depth data at each site. The distribution of peat depth data was based on manual probing of peat depth and derived from a wider peat depth database for the Bleaklow and Kinder Scout areas collected by Moors for the Future Partnership. The spacing of these data points at 100 m centers meant that a limited number of points fall within the study catchments. Therefore, representative peat depth distributions were derived for the local blanket peats



Comparative Data for the Two Study Catchments

	Upper North Grain	Kinder River catchment			
Area (km <sup>2</sup> )	1.05	1.12			
Gully Area (%)	15.1	25.9			
Mean gully depth (m)	1.12	1.21			
Gully length (km)	31.1	39.8			
Bare peat (%)	7.3	Circa 15 (33.5 in 2008)			
Peat Depth (cm)	208	161			

*Note*. Gully data are calculated from gully maps of the catchment derived from the method of Evans and Lindsay (2010) based on 2 m resolution LiDAR-derived digital elevation data. Bare peat areas (areas where erosion has removed all surface vegetation) are derived from vegetation mapping based on a supervised classification of R-G-B-NIR aerial photography developed by Liddaman (2009). The lower figure for Kinder River incorporated revegetation work undertaken since 2,010. Peat depth data are the average of peat probe data for each catchment.

by incorporating additional sample points (within 2 km of the catchment) on Alport Moor (Upper North Grain) and the wider Kinder Scout Plateau (Kinder River) were included. For Kinder Scout, n = 43 and for Upper North Grain, n = 45. Peat depths from the Kinder River site were lower on average and more variable, consistent with extensive surface erosion in the catchment. On Kinder Scout, although gully erosion was extensive, there were also large expanses of bare peat on the interfluves (approximately 33% of the catchment: Table 1). These are areas where the surface vegetation has been completely removed and where considerable sheet erosion has occurred. As such, these areas of bare peat represented "fossil" surfaces where the surface peat was older than the surficial peat in the remainder of the catchment. In contrast, the Upper North Grain catchment was dominated by extensive gully erosion but the interfluves remained vegetated and surface erosion were minimal. Further comparative data on erosion in the two catchments are given in Table 1.

## 2.2. Derivation of Age-Depth Curve for South Pennine Peatlands

To use <sup>14</sup>C ages of DOC as a fingerprint of the locus of in the peat profile, an estimate of the average age-depth relation for catchment peats was required. Age and depth data were compiled from an extensive body of

existing paleoecological data from the Bleaklow and Kinder Scout Plateaux (Conway, 1954; Livett et al., 1979; Tallis, 1964, 1994, 1995; Tallis & Livett, 1994; Tallis & Switsur, 1983, 1990). From these published data, 35 radiocarbon dates were derived and further age-depth data were obtained by combining these dates with established core correlations (based on pollen zoning) in the original literature. In this way, 128 separate dated horizons were identified in 43 peat cores from the Bleaklow and Kinder Scout plateaux. Radiocarbon dates (and associated error bounds) were calibrated using OxCal 4.2.4 and the IntCal 2,013 curve (Reimer et al., 2013). Linear regression was performed to create the age-depth curve.

### 2.3. Radiocarbon Dating of DOC

Water samples for radiocarbon analysis were collected between March 2012 and May 2014 in Nalgene bottles, which had been acid washed in the laboratory and then further washed three times in the streamwater before taking the samples. Samples were collected whilst in the falling stages of high flow events (sampling from the Kinder River on 6/9/13 and 12/5/14 and from Upper North Grain on 4/3/12, 15/5/13 and 7/3/14). Samples were filtered through pre-ashed Whatman GF/F filter papers and submitted to either the Natural Environment Research Council radiocarbon facility or to the Chrono laboratory at Queens University Belfast for dating of DOC. In each case, samples were analyzed to determine the age of the particulate organic carbon (POC) retained on the filter papers.

Assigning a single "date" to DOC samples from peatlands is problematic since measurements pertain to a mixture of material of differing ages contributed from across the peat profile. Evans, Page, et al. (2014) developed a new approach to generating ages from <sup>14</sup>C data in these systems. This approach assumed that rather than representing a single "date," <sup>14</sup>C dates of DOC are the sum of a weighted average of the <sup>14</sup>C contribution of each year represented in the peat profile. The weighting assumed an exponential decline in the proportion of total DOC derived from each year class, which was justified on the basis that both peat decomposition rate (the source of the aged carbon) and hydraulic conductivity (controlling flux of soluble products from peatland soils) typically exhibit exponential decline with depth in peatland soils (Morris et al., 2015). DOC age is calculated as in Equation 1:

$$DO^{14}C = \sum_{t=1}^{t=10,000} \left( {}_{14}CO_2t \times \exp^{(-kt)} \right)$$
(1)

where  ${}^{14}\text{CO}_2$ t is the atmospheric  ${}^{14}\text{C}$  concentration in year *t* and the constant "k" is derived iteratively to fit the measured  ${}^{14}\text{C}$  % modern value from the sample. The  ${}^{14}\text{C}$  % modern contribution post-1950 was atmospheric

concentrations, whereas pre-1950, they were derived from the Intcal13 radiocarbon calibration curve (Reimer et al., 2013).

The Evans, Page, et al. (2014) approach was applied to determining the "age" of DOC from the study sites. The fitting process can produce two solutions for *k* for samples containing a substantial proportion of modern (post 1950s) material, and following Evans, Page, et al. (2014), the older date was used in these degraded systems. This method and the derived ages are subsequently referred to as Model 1, which takes a default approach and assumed that the full peat profile was present in the catchments. Model 1 was therefore used to calculate DO<sup>14</sup>C for both the Upper North Grain and Kinder River catchments. However, on Kinder Scout, extensive surface erosion means that this approach may not be accurate since the youngest peats may have been eroded from the catchment. The Evans, Page, et al. (2014) model was therefore adjusted for this site to allow for calculation of DOC ages based in areas where the old peat was exposed at the surface (Model 2).

The range of peat depths across the two sites (Figure 3: 0–350 cm) was consistent, which is to be expected given comparable basal ages and climate. Therefore, the reduced mean peat depth and increased variability observed on Kinder Scout may be reasonably assumed to be due to erosion of surficial peats observed in this catchment. In Upper North Grain, erosion was confined to gullies, and gully sites were not included in the peat depth measurements. The difference in mean depth between sites was 0.47 m, which was interpreted to represent the average depth of surface erosion in the Kinder catchment. The Model 2 version of the Evans, Page, et al. (2014) model was therefore modified for the Kinder catchment to model DO<sup>14</sup>C accounting for areas of surficial erosion in some areas of the catchment as follows.

The surface erosion in the Kinder River catchment was concentrated within bare peat areas, which represented 33% of the catchment area (Table 1); therefore, the depth of peat removed from these sites may be taken as 1.42 m (1/0.33  $\times$  0.47 m), assuming that erosion is negligible on vegetated surfaces and so the catchment averaged erosion is concentrated on the bare peat locations. This depth was used in combination with the age depth relationship detailed in the results to calculate the likely age of surface peats in eroded areas, which was 3,577 years. Model 2 calculates the estimated age based on carbon sourced from two theoretical peat profiles. The first eroded peat profile (i.e., excluding years 0–3,577) represents the eroded third of the catchment, and the second intact peat profile represents the remaining two thirds of the catchment (excluding gullied areas where the complete peat profile has been removed). Model 2 iteratively calculates a revised single value for the exponent *k* (determining the depths from which DOC is sourced) assuming a 67% contribution of carbon from the intact peat profile and a 33% combination from the eroded profile.

Equation 1 remains the same but it is calculated separately for two peat columns and the results combined weighted by area (33% (eroded catchment) and 67% (uneroded catchment)) as indicated by Equation 2.

$$\mathrm{DO}^{14}C = \left(\sum_{t=1}^{t=10,000} \left(_{14}\mathrm{CO}_2 t \times \exp^{(-kt)}\right) \times 0.67\right) + \left(\sum_{t=3578}^{t=10,000} \left(_{14}\mathrm{CO}_2 t \times \exp^{(-kt)}\right) \times 0.33\right)$$
(2)

The two versions of this model (1 for UNG and 1 and 2 for Kinder) were used to produce a single average DOC age for the study sites. The Evans, Page, et al. (2014) approach to determining the age of DOC in waters means that in addition to an estimated mean DOC age, an age distribution can be produced, which effectively represents the ages of the layers of decomposing peat thought to be contributing DOC to the measured waters. In this study, the age depth curve derived for the Bleaklow and Kinder Scout region allowed the conversion of these age distributions to depth distributions. This approach provided a basis for estimating the relative contribution of near-surface peat layers to DOC export.

The approach to modeling age of DOC rests on several assumptions. The core assumption of the Evans, Page, et al. (2014) model that the contribution of DOC from deeper layers declines exponentially is well supported by conventional understanding of vertical changes in hydraulic conductivity and decomposition in peatlands. Model 2 further assumes that we have accurately estimated the area of bare peat and the removal of surface peats in this area. In order to test these assumptions, a sensitivity analysis was undertaken to assess the degree to which error in these parameters would influence our results.

Model 2 was calculated 36 times varying both bare peat area (25%-50%) in increments of 5%) and surficial erosion depth (20–60 cm in 10 cm increments). This led to a total of 36 iterations of Model 2. The model results



Table 2   Data Used for Water Table Model Testing						
Date range	Measurements	n (wells per site)	Number of sites			
11/06–10/08	23 (monthly)	3	2 sites			
7/07–11/08	16 (monthly)	1–3	2 sites			
2/11-12/13	35 (weekly September–December)	15	1 site			
9/10-12/10 and 9/11-12/11	20 (weekly)	3–5	2 sites			

were interpolated using the Origin software package to visualize the range of age outputs across this range of input variables in order to assess the potential error associated with these parameters. Model 2 also assumes that the eroded and intact areas contribute carbon in proportion to their areas.

## 2.4. Water Table Modeling

An empirical spatial model of median water table across the Bleaklow and Kinder Scout plateaux was derived based on regression of the relationship between topographic wetness index (WI) (calculated as the average WI value in a 30 m square based on a 2 m resolution LiDAR Digital Elevation Model (DEM), where WI =  $\ln (a/s)$ , a = upslope contributing area and s = slope) and median water table, measured at 12 sites across Kinder Scout and Bleaklow plateau (Allott et al., 2009). Water table was measured manually as depth below the peat surface in 15 dipwells (See Allott et al., 2009 for detailed methodology) in a 30 m diameter plot concurrently at each of these sites every two weeks between April and November 2008 (9 measurement intervals). Median water table for each site was derived by taking the mean of the 15 measurements at each site and further calculating the median site value across the nine measurement intervals.

Allott et al. (2009) reported water table drawdown in heavily gullied peats on the Bleaklow and Kinder Scout Plateaux. These results demonstrate that this was a very local phenomenon with drawdown limited to a 2 m distance from gully edges. As a consequence, all WI model training and test data were from vegetated interfluve sites away from gully edges. In order to incorporate gully edge effects into the empirical water table model, an additional 200 mm of water table depression was modeled for a 2 m (one pixel) buffer (Allott et al., 2009) around the gully system. Gully extent was derived from the gully maps presented by Evans and Lindsay (2010).

In order to validate the empirical model, it was tested against an additional independent water table data set as outlined in Table 2. The test data set is comprised of available data compiled from dipwells on the Bleaklow and Kinder Scout Plateau used for other projects. As such, it contains data from sites with different numbers of dipwells and with data collected at varying intervals and over changing time periods, which would introduce spatial and temporal variability into the WI versus. water table relationship. For this reason, these data were considered to provide a robust and conservative test of the empirical model.

The model was used to calculate average median water table depths for each catchment. The Kinder River catchment was modeled in two parts, thus taking into account the different types of erosion in the catchment; (a) areas where the surface was intact (vegetated areas with gully presence 67% of the catchment) and (b) combined gully and surface erosion (bare peat areas, 33% of the catchment).

## 3. Results

#### **3.1. Radiocarbon Ages**

<sup>14</sup>C determinations for all DOC samples, and associated age estimates, are detailed in Table 3. The data from the Kinder River catchment show a tight clustering of ages between sampling sites and on for sampling periods with all samples showing aged DOC (range 88.6%–89.1% modern). This consistency supports the view that DOC age was controlled by the peatland catchment characteristics and that it was not a function of sample location downstream of the peatland catchment. Stimson et al. (2017) also report further data from a site downstream, which confirm these patterns but which are excluded from this analysis because of an intervening tributary.

Radiocarbon Determinations for Dissolved Organic Carbon From Kinder River and Upper North Grain (UNG)					
Publication code	Site	Date	$^{14}C$ enrichment % modern ± 1s	Radiocarbon age (years BP)	Modeled ages (years before sampling date) kinder (model 1/Model 2)
Upper North Grain	catchment				
UBA-20265	UNG1	4/3/12	$99.26 \pm 0.32$	$60 \pm 26$	393
UBA-20266	UNG3	4/3/12	$97.94 \pm 0.33$ $167 \pm 27$		467
UBA-23192	UNG1	15/5/13	$102.58 \pm 0.29$ Moder		247
UBA-23193	UNG2	15/5/13	$103.11 \pm 0.34$	Modern	228
UBA-23194	UNG3	15/5/13	$104.27 \pm 0.39$	Modern	193
UBA-25463	UNG1	7/3/14	$96.51 \pm 0.43$	$285 \pm 36$	557
UBA 25464	UNG2	7/3/14	$97.17 \pm 0.41$	$231 \pm 34$	514
UBA-25465	UNG3	7/3/14	$97.36 \pm 0.34$	$215 \pm 28$	502
Kinder River catch	ment				
SUERC-50395	Kinder1	06/9/13	$89.06 \pm 0.41$	931 ± 37	1,178/1,441
SUERC-50396	Kinder2	06/9/13	$88.61 \pm 0.40$	972 ± 37	1,223/1,463
SUERC-54801	Kinder1	12/5/14	$88.56 \pm 0.41$ 976 ±		1,228/1,466
Note DD Defens D	hacant				

Table 3

Note. BP: Before Present

For Upper North Grain, a greater degree of variability between samples was observed. All samples from this catchment showed a greater degree of <sup>14</sup>C enrichment (range 96.5%–104.3% modern) than any of the samples collected from the Kinder catchment. Four of the samples returned modern radiocarbon ages, three of which related to a single sampling visit. There was no consistent spatial variation downstream in <sup>14</sup>C between the sampling sites within the catchment (Table 3).

Mean conventional radiocarbon ages (i.e., assuming that samples contained material of uniform age) for DOC in waters from the Kinder River site were 960  $\pm$  62 BP (n = 3) and from Upper North Grain 96  $\pm$  116 years BP (n = 8). Error ranges on these estimates were 95% confidence intervals of the mean age.

## 3.2. Peat Age Depth Curves

The age depth curve derived from published radiocarbon dates and correlated horizons on the Bleaklow and Kinder Scout plateaux is presented in Figure 4. Across the whole region, simple linear regression of the radiocarbon ages against centroid of depth produces a linear age-depth model over Holocene timescales with the age and peat depth relationship being D = 0.03720 A ( $R^2 = 0.91$  P < 0.0001), where D is peat depth in cm and A is calibrated radiocarbon age in years. The data density was much greater in the most recent part of the Holocene. Multiple points representing the same age are indicative of varying depths of correlated horizons found in multiple cores.

#### 3.3. Modeled Radiocarbon Ages

Application of the Evans, Page, et al. (2014) approach to determining age (Model 1), which considers DOC as a continuous mixture of ages, yielded older dates with average ages of 1,210 years BP for the Kinder River catchment and 388 years BP for Upper North Grain.

Model 2 calculated an average age of DOC from the Kinder River catchment of 1,457 years BP.

Mean radiocarbon levels and estimated ages for POC retained on filter papers are shown in Table 4. All POC samples had pre-bomb <sup>14</sup>C levels, and all were <sup>14</sup>C-depleted compared to DOC samples. Average radiocarbon ages were 2,761 (Kinder River n = 1) and 2,842 ± 507 (Upper North Grain n = 5). The model of Evans, Page, et al. (2014) was not applied because the underlying assumptions regarding the age distribution of material in the samples were unlikely to be valid for POC generated by spatially heterogeneous erosion processes.



## Journal of Geophysical Research: Biogeosciences



**Figure 4.** Peat Age-Depth relations for the Bleaklow Plateau and Kinder Scout area. Error bars for peat depth were not quantified. Error bars for age are the errors from OXCAL. The solid line represents the linear regression for the mean age produced from OXCAL; the shaded area is the standard error of the estimate from the linear regression ( $\pm$ 24 cm). Original data from: Conway, 1954; Livett et al., 1979; Tallis, 1964, 1985, 1995; Tallis & Livett, 1994; Tallis & Switsur, 1983, 1990.

#### 3.4. Modeling Source Depths of DOC

Figure 5 shows the estimated age distributions of DOC from the Evans, Page, et al. (2014) model, and those plots converted to depth distributions on the basis of the age depth relationships (Figure 4). Figure 5a describes the age and depth relationships at Upper North Grain. Figure 5b shows the age and depth relationships in the Kinder River catchment utilizing Model 1 (not accounting for surface erosion). Figure 5c presents the second approach (Model 2) to account for surface erosion in the Kinder River catchment, assuming that surface erosion was concentrated in one third of the catchment. The age profiles for Kinder River Model 2 are presented on two different scales (0–5,000 and 3,577–8,577), the former for the intact part of the catchment and the latter for the area of concentrated surface eroded section of the catchment. These curves are identical as k (Equation 1) was recalculated to take into account both areas of the catchment in their appropriate proportions.

The mean lower quartile, 50th percentile, and upper quartile are marked on Figure 5. Standard error of the mean was calculated for each quartile across the sites. The lower quartile, 50th percentile, and upper quartile, respectively, were at approximately 106 ( $\pm$ 24), 237 ( $\pm$ 42), and 497 ( $\pm$ 42) years, or 4 ( $\pm$ 0.80), 9 ( $\pm$ 1.46) and 19 ( $\pm$ 2.09) cm in the UNG catchment, whereas in the Kinder River catchment (Model 1), they were 358 ( $\pm$ 5), 822 ( $\pm$ 11), and 1,697 ( $\pm$ 23) years, or at 13 ( $\pm$ 0.17), 31 ( $\pm$ 0.42), and 63 ( $\pm$ 0.87) cm. The results clearly suggested that DOC was sourced from a greater range of depths in the Kinder River catchment Model 1 than in the Upper North Grain catchment. However, Kinder River Model 2 suggests source depths that are more comparable with UNG with

Radiocarbon Determinations for Particulate Organic Carbon (POC) Samples From Kinder River and Upper North Grain (UNG)						
Publication code	Site	Date	$^{14}$ C enrichment % modern ± 1s	Radiocarbon age (years BP)		
UBA-20262	UNG1	4/3/12	$76.64 \pm 0.28$	2,137 ± 29		
UBA-20263	UNG3	4/3/12	$69.71 \pm 0.26$	$2,961 \pm 30$		
UBA-23189	UNG1	15/5/13	$67.26 \pm 0.26$	3,187 ± 31		
UBA-23190	UNG2	15/5/13	$68.71 \pm 0.24$	$3,014 \pm 29$		
	Radiocarbon Determin. (UNG) Publication code UBA-20262 UBA-20263 UBA-23189 UBA-23190	Radiocarbon Determinations for Particula(UNG)SitePublication codeSiteUBA-20262UNG1UBA-23189UNG1UBA-23190UNG2	Radiocarbon Determinations for Particulate Organic Carl (UNG)Publication codeSiteDateUBA-20262UNG14/3/12UBA-20263UNG34/3/12UBA-23189UNG115/5/13UBA-23190UNG215/5/13	Radiocarbon Determinations for Particulate Organic Carbon (POC) Samples From Kinder Ri (UNG)Publication codeSiteDate $^{14}C$ enrichment % modern ± 1sUBA-20262UNG14/3/1276.64 ± 0.28UBA-20263UNG34/3/1269.71 ± 0.26UBA-23189UNG115/5/1367.26 ± 0.26UBA-23190UNG215/5/1368.71 ± 0.24		

 $67.72 \pm 0.25$ 

 $70.91 \pm 0.37$ 

15/5/13

12/5/14

Note. BP: Before Present.

UNG3

Kinder1

UBA-23191

SUERC-54372

Table 4

 $2,913 \pm 30$  $2,761 \pm 41$ 



2169896, 2022, 7, Downloaded from https://appubs.onlinelibary.wiley.com/doi/01.029/02110306344 by Ukri CO U.S.hared Business NATURAL ENVIRONMENT RSCH COUNCIL, Wiley Online Library on [02/11/2022] See the Terms and Conditions (https://onlinelibary.wiley.com/term

and-conditions) on Wiley Online Library

for rules of use; OA a



**Figure 5.** Modeled source ages and equivalent depths of dissolved organic carbon for all models. 25th, 50th, and 75th percentiles marked as dotted lines. The black line represents the mean and the colored lines represent each individual sample. There were 8 samples at Upper North Grain (UNG) and 3 at Kinder River. (a and b) UNG catchment; (c and d) Kinder River catchment Model 1; (e–g) Kinder River catchment Model 2. Age distributions presented both assuming that the surface begins at 0 years (e) and at 3,577 years (on the basis of calculations of the age of surface erosion as in Section 3.3). Error bars on the peat depth frequency plots represent one standard error of the estimate from Figure 4.

the lower quartile, 50th percentile, and upper quartile being  $3,652 (\pm 2), 3,756 (\pm 6), \text{ and } 3,933 (\pm 11)$  (equivalent to 74, 179, and 358 years older than the surface date) or  $3 (\pm 0.09), 7 (\pm 0.21)$ , and  $13 (\pm 0.41)$  cm, respectively. Kinder River Model 1 implies peat loss throughout the profile, while Kinder River Model 2 has peat loss confined to the upper layers of the peat profile in common with UNG.





Figure 6. Relation between wetness index and median water table depth at 12 vegetated sites across the Kinder Scout and Bleaklow Plateau. The dashed line is line of best fit and the solid line is the 1:1 line.

#### 3.5. Water Table Modeling

Figure 6 shows the relation between site WI derived from 2 m DEM data and median water table for 12 sites across Bleaklow and Kinder Scout. The empirical relationship implied that beyond WI values of around 5, median water table is approximately constant. This is consistent with physical understanding of peatland systems with a thin upper acrotelm layer of very high hydraulic conductivity where water table residence times are short (Holden & Burt, 2003b). The empirical model was validated against available water table data from other sites across Bleaklow and Kinder Scout and Figure 7 shows observed and predicted values. The test data are from smaller dipwell arrays and for a range of sample collection periods. This is a very conservative test of the model (as both conditions are likely to increase variability in water table relative to WI due to synoptic variability and the variability of peatland water tables; Allott et al., 2009). In this context, the model performance ( $R^2 = 0.64$  SE 100.8 and the mean estimated = 0.89) gives confidence that the model adequately represents long-term average water table positions.



**Figure 7.** Observed and predicted water table for validation sites. The black regression line relates to the test data set only.

Modeled median water table for the two study catchments is presented in Figure 8. Mean median water table depth in Upper North Grain was 194  $\pm$  45 mm and in the Kinder River catchment, it was 348  $\pm$  45 mm. To complement the "Model 2" modeling of DOC ages for the Kinder catchment, water tables for Kinder Scout were also calculated separately for areas of surface erosion (~33% of the catchment) and vegetated areas (~67% of the catchment). These were modeled as 340 and 358 mm, respectively. Figure 9 presents the cumulative frequency distribution of water table depths in the two catchments indicating that near-surface water tables (<150 mm) were preserved in over 50% of the Upper North Grain catchment, whereas in the Kinder River catchment, this value was approximately 20%. The modeling of water table depths allows comparison of catchment average water table conditions with the depths from which the <sup>14</sup>C models suggest that DOC is being sourced.

#### 3.6. Comparing Water Table Models and DOC Source Depths

Figure 10 compares the modeled mean 50th percentile source depths of DOC within the peat derived from the analysis in Figure 5 to mean modeled water table depth for the two catchments, derived from the analysis presented in Figure 8. To allow the comparison of the two Kinder River models, the water







table depths presented are from the gullied (i.e., relating to Model 1) and gullied and surface eroded (i.e., relating to Model 2) portions of the catchment.

The modeled water table depths were consistently deeper than the modeled average <sup>14</sup>C depth although this was less apparent for Kinder Model 1. For both sites model 1, <sup>14</sup>C depth lies in the predicted range of water table although the median modeled water table was within the interquartile error range of the DOC data only for Kinder Model 1. There was a much larger discrepancy between the modeled water table depth and the modeled average <sup>14</sup>C depth for Kinder River Model 2.

## 3.7. Sensitivity Analysis

The influence of bare peat area and erosion depth assumptions on model 2 output is presented in Figure 11. Increasing both erosion depth and bare peat area increases the model age output linearly.



Figure 9. Modeled distributions of water table for the two study catchments.

Modeling  $a \pm 25\%$  proportional error in erosion depth (35–59 cm) and error in bare peat coverage (25%–41%) resulted in age outputs of 1,288–1,696 years BP and 1,369–1,564 years BP, respectively. The age outputs from the model are relatively insensitive to minor error in these parameters.

## 4. Discussion

#### 4.1. Radiocarbon Age of DOC in Runoff

The data presented here represent the most extensive suite of radiocarbon ages of DOC reported to date from heavily eroded peatlands in the UK. For both catchments studied, there was limited variability in DOC ages between sampling sites and no consistent change in DOC ages downstream. This result supports the idea that the catchment sources, rather than in-stream carbon transformations, govern the radiocarbon age of DOC at the headwater catchment scale. In the Upper North Grain catchment, younger (modern) ages were obtained from one sampling event, suggesting some temporal variability in the source of DOC as a function of hydrological conditions. This is consistent with other studies that have shown previous UK blanket bog studies that have shown higher levels of <sup>14</sup>C enrichment in DOC during periods







of higher flow, when more DOC is sourced from shallower horizons (Evans et al., 2007; Tipping et al., 2010). However, higher resolution sampling would be required to substantiate this finding at the study sites used here.

Previously reported DOC dates from comparatively undisturbed UK peatlands have largely indicated modern or near modern ages, indicative of turnover of litter and decomposition of recently deposited peat (Billett et al., 2011; Evans, Page, et al., 2014; Palmer et al., 2001; Tipping et al., 2010). These dates are consistent with aerobic decomposition in the acrotelm layers of relatively intact peatland systems. In contrast, two dates on eroded southern Pennine systems have indicated older sources of carbon with conventional radiocarbon ages of 463 and 1763 years BP (Evans, Page, et al., 2014). The three dates from the Kinder River catchment presented here have produced age estimates from 931 to 976 years BP, which support the view that the single extreme date from Evans, Page, et al. (2014) is not an outlier. The data support the idea that in degraded UK



peatlands, older <sup>14</sup>C ages indicative of decomposition of deeper peat layers are characteristic. These systems are therefore losing carbon at depth in a similar manner (although not to a similar degree) to the pattern reported for tropical peatlands in Evans, Page, et al. (2014) and for drained boreal agricultural peatlands by Hulatt et al. (2014). Eight dates from the less eroded Upper North Grain catchment ranged from modern to 285 years BP, and one included in Evans, Page, et al. (2014) (95.11% modern or ~403 years BP), similarly indicate decomposition of deeper peat layers, but to a lesser degree.

Applying the Evans, Page, et al. (2014) model resulted in the average modeled dates being consistently older than the average measured dates for both UNG (measured: 106; modeled: 388 years BP) and Kinder (measured: 1,014; modeled 1 and 2: 1,210/1,457 years BP). Since DOC is almost always composed of a broad range of compounds with variable ages, this age-continuum model likely provides a more realistic mean age estimate than the single-year conventional age model.

#### 4.2. Controls on DOC Age in Degraded Peat Systems

**Figure 11.** Sensitivity analysis for the major assumptions of Model 2. Contour analysis based on the age outputs of 36 iterations of Model 2. The age output from Model 2 of 1,457 years Before Present (BP) is marked with an X and other iterations as dots.

One potential control in DOC ages measured in stream systems is in stream transformation of carbon (e.g., from POC: Brown et al., 2019). In the catchments studied here, POC ages were consistently much older than DOC ages.

PO<sup>14</sup>C average ages were 2,761 (Kinder River n = 1) and 2,842 ± 507 (Upper North Grain n = 5). The similarity in age between catchments indicates that POC to DOC transformation is not the explanation for varied DOC ages between catchments. The small catchments are of comparable scales and stream residence times in stormflow conditions are likely to be on the order of 20 min, so that in-stream DOC is expected to reflect carbon source rather than in-stream processes.

There is a distinct contrast in the age of DOC draining from the two study catchments with aged DOC being produced from both systems but older DOC from Kinder Scout. The <sup>14</sup>C modeling results allow us to address the three hypotheses relating erosion and the production of older DOC.

- 1. The water table data and the distributions of depth of peat decomposition derived in this study (Figure 10) allow us to assess the role of water table changes as drivers of deep peat decomposition in these degraded systems. Water table and predicted source depths for DOC were closer to the surface in Upper North Grain (Figure 10). More intense dissection of the Kinder Scout plateau produced lower median water tables (Figures 8 and 9) across the headwater peatlands of the Kinder River, and measured DOC ages indicated DOC sourced from deeper in the peat at this location when modeling the catchment using model 1 (assuming there had been no surface erosion) (Figure 10). For both catchments, the predicted DOC source depths were within the range of predicted water table depths. This result was consistent with water table control on DOC production (Figure 1a).
- 2. Where extensive erosion has removed vegetation so that bare peat is exposed at the surface, the supply of fresh DOC from litter is reduced. We can assess the relative importance of this mechanism (Figure 1b) with reference to Figure 5. Across both sites (UNG and Kinder Model 2), the proportion of DOC, which was derived from the material aged 0–9 years, was estimated at just 1.8%–5% so that the reduced supply of modern DOC is unlikely to be the primary cause of differences in the age of DOC exported from the two catchments.
- 3. Where there has been extensive surface erosion of bare peat, or fire-related peat loss (Evans, Page, et al., 2014), then much older peat from lower down the profile will be exposed at the surface, and DOC may be sourced from the older peat material that now lies within the upper aerobic zone (Figure 1c). In the Kinder River catchment, there is extensive bare peat. Model 2 that adapted the Evans, Page, et al. (2014) model to account for surface erosion in one third of the catchment predicted sourcing of DOC from a shallow zone (75% DOC from above 14 cm) at the surface of the peat profile, which was not consistent with the modeled water table depth (Figure 10). This is in contrast with the Upper North Grain results where the zone of production of the majority of DOC appeared to be within the range of the modeled WT. Applying the original Evans, Page, et al. (2014) model (Model 1) to the Kinder River data yielded a more widely distributed depth distribution of DOC production than UNG with some DOC production occurring in deeper peats consistent with deeper water tables modeled for this catchment. However, Model 1 did not account for surface erosion. Model 2 explicitly accounted for measured surface erosion but counterintuitively predicted DOC production limited to the shallow near surface. Model 2 assumed old surface peat across 33% of the catchment (circa 3.5ka). If DOC were sourced significantly from deep peat in these areas, a much older date than the observed <sup>14</sup>C date of  $960 \pm 62$  years BP would be expected. The dominant control of erosion hypothesized in Figure 1c is therefore not supported.

Both the theory and the data from Upper North Grain support the idea that water table was controlling decomposition of the peat. The mismatch of water table and the modeled source depths for DOC in Model 2 for the Kinder Catchment suggested that in this case the assumptions of the modified Evans model break down. The sensitivity analysis indicated that the discrepancy between predicted and actual DOC age is unlikely to be driven by error in the erosion depths or percentage bare area, which were inputs to model 2. The implication therefore is that the DOC flux from older peats in areas of surficial erosion was reduced, compared to intact peat profiles, impacted by gully erosion. The apparent congruence of water table and DOC source depths predicted by model 1 suggests that the uneroded sites dominate DOC production.

Two possibilities arise: (a) that rates of decomposition, and so, of DOC production in eroded areas are reduced because the surficial peat is older and less labile or; (b) alternatively that the flow pathways in heavily eroded systems lead to reduced interaction of runoff and the peat matrix. Bare peat respiration rates at nearby sites in the Peak District have been shown to be low and show no seasonal change (Caporn et al., 2007), supporting hypothesis



one. The second hypothesis is consistent with some evidence from drained systems that macropore flow (bypassing the peat matrix) is much more extensive in peatlands with lowered water tables (Holden et al., 2006).

#### 4.3. Carbon Age and Carbon Flux

Dissolved carbon ages from both catchments indicate loss of deep peat such that it might be expected that absolute dissolved carbon fluxes from the systems are high. Previous research in the catchments supports this view. Pawson et al. (2012) report DOC flux from UNG of 27.5 MgC km<sup>-2</sup> a<sup>-1</sup>. For the Kinder River catchment, Stimson et al. report a flux of 8.6 MgC km<sup>-2</sup> a<sup>-1</sup> for the 3.9 km<sup>2</sup> catchment above the Kinder reservoir. However, only the upper portion of this catchment is peat covered (the 1.1 km<sup>2</sup> catchment considered in this study). If it is assumed that the DOC is sourced from the peat covered catchment, this gives a DOC flux of 29.9 MgC km<sup>-2</sup> a<sup>-1</sup>, which would be a maximum estimate for this system. These values exceed typical peatland fluvial DOC flux values reported by Aitkenhead and McDowell (2000) (5.7-8.6 MgC km<sup>-2</sup> a<sup>-1</sup>) and also values reported for eroded headwater peats in northern England (Worrall et al., 2006) (13.4–22.5 MgC km<sup>-2</sup> a<sup>-1</sup>). These results are therefore consistent with elevated dissolved carbon flux from degraded peats with carbon sourced from deep peat sources. However, the significantly older carbon derived from the Kinder River catchment is not matched by increased rates of carbon loss. It is possible that in extremely degraded systems, such as the Kinder Plateaux, reduced rates of microbial activity (implied by reduced soil respiration on bare peat such as seen at some Finnish peat extraction sites; Shurpali et al., 2008) and modifications of hydrological pathways act to moderate DOC loss. These results are relevant to other areas of expansive bare peat, such as extensive peat mining (e.g., Haghighi et al., 2018; Salm et al., 2012).

## 5. Conclusions

We observed deep carbon loss from degraded temperate peatland systems comparable to the results reported for a wider set of global peatlands by Evans, Page, et al. (2014). These UK upland peatlands are losing carbon that has been stored for timescales in excess of a millennium. A new approach to modeling the source of DOC has demonstrated that old DOC was sourced from greater depths in the peat profile, and the similarity of measured water tables to the areas of the peat profile, which is the source of this DOC, was consistent with water table drawdown as the primary control on DOC ages. The removal of the youngest layers of peat by surface erosion appeared to have a smaller than expected impact on the age of carbon exported from the system. This discrepancy needs further investigation, but potentially suggests that either (a) bare eroded peat is less biologically active due to a combination of the presence of recalcitrant materials and lack of plant mediated processes and input of fresh material to drive microbial processes or (b) the recognized expansion of macropore density in drained peats may be a limiting factor on deep carbon loss since a proportion of throughflow bypasses the peat matrix limiting DOC export.

This paper has demonstrated a clear link between gully erosion of peatlands, reductions in water table, and the loss of old carbon stored in the peat soils to the fluvial system. In these eroded peatlands, the form of peatland erosion appeared to have a significant impact. Gully erosion appeared to be more significant than surficial erosion as a factor in the loss of old DOC from eroded peats.

Across the globe, peatland drainage and degradation are widespread (Bonn et al., 2016) and threaten long-term carbon storage. A growing body of evidence indicates that loss of old organic carbon from peatlands is an indicator of deep decomposition linked to degradation. The approaches developed here if applied more widely at small catchment scales have the potential to explore the links between degradation and deep carbon loss across a range of peatland types. Similarly, the aim of many peatland restoration projects is to raise water table in order to reduce carbon losses from the peat store. The association between radiocarbon age of DOC and water table indicated here suggests that <sup>14</sup>C dating of DOC might be a useful catchment scale proxy for water table, a direct measurement of the locus of carbon loss from the peat, and an effective measure of the overall "health" of the peatland ecosystem. The approach therefore has potential as a rapid method of assessing stabilization and restoration of intact peatland function at landscape scale in comparison to intensive and expensive monitoring campaigns.

This paper has exploited the correlation between peat depth and peat age, which along with new approaches to interpreting radiocarbon ages of peatland waters has allowed us to identify where in the depth profile dissolved carbon is being sourced. There is a long history of study of the chronostratigraphy of peatlands, which can



potentially be utilized, together with new <sup>14</sup>C data on peatland waters to explore the interactions of peatland hydrology and carbon loss from these important terrestrial carbon stored.

## **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

## **Data Availability Statement**

Data sets for this research are available here https://doi.org/10.5281/zenodo.4548224.

## References

- Aitkenhead, J. A., & McDowell, W. H. (2000). Soil C:N ratio as a predictor of annual riverine DOC flux at local and global scales. *Global Biogeochemical Cycles*, 14(1), 127–138. https://doi.org/10.1029/1999GB900083
- Alderson, D. M., Evans, M. G., Shuttleworth, E. S., Pilkington, M., Spencer, T., Walker, J., & Allott, T. E. H. (2019). Trajectories of ecosystem change in restored blanket peatlands. Science of the Total Environment, 665, 785–796. https://doi.org/10.1016/j.scitotenv.2019.02.095
- Allott, T. E., Evans, M. G., Lindsay, J. L., Agnew, C. T., Freer, J. E., Jones, A., & Parnell, M. (2009). Water tables in Peak District blanket peatlands (report 17). Moors for the Future.
- Billett, M. F., Garnett, M. H., Dinsmore, K. J., Dyson, K. E., Harvey, F., Thomson, A. M., et al. (2011). Age and source of different forms of carbon released from boreal peatland streams during spring snowmelt in E. Finland. *Biogeochemistry*, 111(1–3), 273–286. https://doi. org/10.1007/s10533-011-9645-4
- Bonn, A., Allott, T., Evans, M., Joosten, H., & Stoneman, R. (2016). Peatland restoration and ecosystem services: An introduction. In A. Bonn, T. Allott, M. Evans, H. Joosten, & R. Stoneman (Eds.), *Peatland restoration and ecosystem services: Science, policy and practice (ecological reviews)* (pp. 1–16). Cambridge University Press.
- Brown, S. L., Goulsbra, C. S., & Evans, M. G. (2019). Controls on fluvial carbon efflux from eroding peatland catchments. *Hydrological Processes*, 33(3), 361–371. https://doi.org/10.1002/hyp.13329
- Butman, D. E., Wilson, H. F., Barnes, R. T., Xenopoulos, M. A., & Raymond, P. A. (2015). Increased mobilization of aged carbon to rivers by human disturbance. *Nature Geoscience*, 8(2), 112–116. https://doi.org/10.1038/ngeo2322
- Caporn, S., Sen, R., Field, C., Jones, E., Carroll, J., & Dise, N. (2007). Consequences of lime and fertiliser application for moorland restoration and carbon balance (Research report to Moors for the Future). Manchester Metropolitan University.

Charman, D. (2002). Peatlands and environmental change. Wiley.

- Clark, J., Gallego-Sala, A., Allott, T., Chapman, S., Farewell, T., Freeman, C., et al. (2010). Assessing the vulnerability of blanket peat to climate change using an ensemble of statistical bioclimatic envelope models. *Climate Research*, 45(1), 131–150. https://doi.org/10.3354/cr00929
- Clay, G. D., & Evans, M. G. (2017). Ten-year meteorological record for an upland research catchment, near the summit of snake pass in the Peak District, UK. Weather, 72(8), 242–249. https://doi.org/10.1002/wea.2824
- Clymo, R. S. (1984). The limits to peat growth. Philosophical Transactions of the Royal Society of London Series B Biological Sciences, 303(1117), 605-654.
- Conway, V. M. (1954). Stratigraphy and pollen analysis of Southern Pennine blanket peats. Journal of Ecology, 42(1), 117–147. https://doi.org/10.2307/2256982
- Dargie, G. C., Lewis, S. L., Lawson, I. T., Mitchard, E. T. A., Page, S. E., Bocko, Y. E., & Ifo, S. A. (2017). Age, extent and carbon storage of the central Congo basin peatland complex. *Nature*, 542(7639), 86–90. https://doi.org/10.1038/nature21048
- Dean, J. F. (2019). Groundwater dependent ecosystems in arid zones canuse ancient subterranean carbon as an energy source in the local food web. Journal of Geophysical Research: Biogeosciences, 124(4), 733–736. https://doi.org/10.1029/2019JG005089
- Dommain, R., Dittrich, I., Giesen, W., Joosten, H., Rais, D. S., Silvius, M., & Wibisono, I. (2016). Ecosystem services, degradation and restoration of peat swamps in the Southeast Asian tropics. In A. Bonn, T. Allott, M. Evans, H. Joosten, & R. Stoneman (Eds.), *Peatland restoration and ecosystem services: Science, policy and practice* (pp. 253–288). Cambridge University Press. https://doi.org/10.1017/CBO9781139177788
- Evans, C., Allott, T., Billett, M., Burden, A., Chapman, P., Dinsmore, K., et al. (2013). *Greenhouse gas emissions associated with non gaseous losses of carbon from peatlands- Fate of particulate and dissolved carbon (Final report to DEFRA SP1205)*. Centre for Ecology and Hydrology. Retrieved from: http://randd.defra.gov.uk/
- Evans, C. D., Bonn, A., Holden, J., Reed, M., Evans, M., Worrall, F., & Parnell, M. (2014). Relationships between anthropogenic pressures and ecosystem functions in UK blanket bogs: Linking process understanding to ecosystem service valuation. *Ecosystem Services*, 9, 5–19. https:// doi.org/10.1016/j.ecoser.2014.06.013
- Evans, C. D., Freeman, C., Cork, L. G., Thomas, D. N., Reynolds, B., Billett, M. F., et al. (2007). Evidence against recent climate-induced destabilisation of soil carbon from <sup>14</sup>C analysis of riverine dissolved organic matter. *Geophysical Research Letters*, 34(7), 1–5. https://doi. org/10.1029/2007GL029431
- Evans, C. D., Page, S. E., Jones, T., Moore, S., Gauci, V., Laiho, R., et al. (2014). Contrasting vulnerability of drained tropical and high-latitude peatlands to fluvial loss of stored carbon. *Global Biogeochemical Cycles*, 28(11), 1215–1234. https://doi.org/10.1002/2013GB004782
- Evans, C. D., Renou-Wilson, F., & Strack, M. (2016). The role of waterborne carbon in the greenhouse gas balance of drained and re-wetted peatlands. Aquatic Sciences, 78(3), 573–590. https://doi.org/10.1007/s00027-015-0447-y
- Evans, M., & Lindsay, J. (2010). High resolution quantification of gully erosion in upland peatlands at the landscape scale. Earth Surface Processes and Landforms, 35(8), 876–886. https://doi.org/10.1002/esp.1918
- Evans, M., & Warburton, J. (2007). Geomorphology of upland peat: Erosion, form and landscape change. John Wiley & Sons. https://doi. org/10.1002/9780470798003
- Evans, M. G., Warburton, J., & Yang, J. (2006). Sediment budgets for eroding blanket peat catchments: Global and local implications of upland organic sediment budgets. *Geomorphology*, 79(1–2), 45–57. https://doi.org/10.1016/j.geomorph.2005.09.015
- Fraser, C. J. D., Roulet, N. T., & Moore, T. R. (2001). Hydrology and dissolved organic carbon biogeochemistry in an ombrotrophic bog. *Hydrological Processes*, 15(16), 3151–3166. https://doi.org/10.1002/hyp.322

## Acknowledgments

This work was supported by NERC radiocarbon allocation NRCF010001 (allocation number 1657.1012) and by DEFRA grant SP1205. We acknowledge the National Trust and United Utilities for permission to work at the sites.

- Frey, K. E., & Smith, L. C. (2005). Amplified carbon release from vast West Siberian peatlands by 2100. Geophysical Research Letters, 32(9), L09401. https://doi.org/10.1029/2004GL022025
- Gallego-Sala, A. V., Charman, D. J., Brewer, S., Page, S. E., Prentice, I. C., Friedlingstein, P., et al. (2018). Latitudinal limits to the predicted increase of the peatland carbon sink with warming. *Nature Climate Change*, 8(10), 907–913. https://doi.org/10.1038/s41558-018-0271-1

Gorham, E., Lehman, C., Dyke, A., Clymo, D., & Janssens, J. (2012). Long-term carbon sequestration in North American peatlands. *Quaternary Science Reviews*, 58, 77–82. https://doi.org/10.1016/j.quascirev.2012.09.018

- Goulsbra, C., Evans, M., & Lindsay, J. (2014). Temporary streams in a peatland catchment: Pattern, timing, and controls on stream network expansion and contraction. *Earth Surface Processes and Landforms*, 39(6), 790–803. https://doi.org/10.1002/esp.3533
- Haghighi, A., Menberu, M. W., Darabi, H., Akanegbu, J., & Kløve, B. (2018). Use of remote sensing to analyse peatland changes after drainage for peat extraction. *Land Degradation & Development*, 29(10), 3479–3488. https://doi.org/10.1002/ldr.3122
- Harden, J. W., Sundquist, E. T., Stallard, R. F., & Mark, R. K. (1992). Dynamics of soil carbon during deglaciation of the Laurentide ice sheet. Science, 258(5090), 1921–1924. https://doi.org/10.1126/science.258.5090.1921
- Hoag, R. S., & Price, J. S. (1995). A field-scale, natural gradient solute transport experiment in peat at a Newfoundland blanket bog. Journal of Hydrology, 172(1–4), 171–184. https://doi.org/10.1016/0022-1694(95)02696-m
- Holden, J., & Burt, T. P. (2003b). Hydrological studies on blanket peat: The significance of the acrotelm-catotelm model. *Journal of Ecology*, 91(1), 86–102. https://doi.org/10.1046/j.1365-2745.2003.00748.x
- Holden, J., & Burt, T. P. (2003a). Runoff production in blanket peat covered catchments. Water Resources Research, 39(7), 1191–1200. https:// doi.org/10.1029/2002wr001956
- Holden, J., Evans, M. G., Burt, T. P., & Horton, M. (2006). Impact of land drainage on peatland hydrology. Journal of Environmental Quality, 35(5), 1764–1778. https://doi.org/10.2134/jeq2005.0477
- Hulatt, C. J., Kaartokallio, H., Oinonen, M., Sonninen, E., Stedmon, C. A., & Thomas, D. N. (2014). Radiocarbon dating of fluvial organic matter reveals land-use impacts in boreal peatlands. *Environmental Science & Technology*, 48(21), 12543–12551. https://doi.org/10.1021/es5030004
- Joosten, H. (2010). The global peatland CO<sub>2</sub> picture: Peatland status and drainage related emissions in all countries of the world. Wetlands International.
- Liddaman, L. C. (2009). Geospatial patterning and the hydro-geomorphological function of eroding peatland PhD thesis. University of Manchester.
- Lindsay, R. A., Charman, D. J., Everingham, F., O'Reilly, R. M., Palmer, M. A., Rowell, T. A., & Stroud, D. A. (1988). The flow country- the peatlands of caithness and sutherland. Joint Nature Conservation Committee.
- Livett, E. A., Lee, J. A., & Tallis, J. H. (1979). Lead, Zinc and Copper analyses of British blanket peats. Journal of Ecology, 67(3), 865–891. https://doi.org/10.2307/2259219
- Marwick, T. R., Tamooh, F., Teodoru, C. R., Borges, A. V., Darchambeau, F., & Bouillon, S. (2015). The age of river-transported carbon: A global perspective. *Glocal Biogeochemical Cycles*, 29(2), 122–137. https://doi.org/10.1002/2014GB004911
- Moore, S., Evans, C. D., Page, S. E., Garnett, M. H., Jones, T. G., Freeman, C., et al. (2013). Deep instability of deforested tropical peatlands revealed by fluvial organic carbon fluxes. *Nature*, 493(7434), 660–663. https://doi.org/10.1038/nature11818
- Morris, P. J., Baird, A. J., & Belyea, L. R. (2015). Bridging the gap between models and measurements of peat hydraulic conductivity. Water Resources Research, 51(7), 5353–5364. https://doi.org/10.1002/2015wr017264
- Ostle, N. J., Levy, P. E., Evans, C. D., & Smith, P. (2009). UK land use and soil carbon sequestration. Land Use Policy, 26(S1), S274–S283. https://doi.org/10.1016/j.landusepol.2009.08.006
- Page, S. E., & Baird, A. J. (2016). Peatlands and global change: Response and resilience. Annual Review of Environment and Resources, 41(1), 35–57. https://doi.org/10.1146/annurev-environ-110615-085520
- Page, S. E., Rieley, J. O., & Banks, C. J. (2011). Global and regional importance of the tropical peatland carbon pool. *Global Change Biology*, 17(2), 798–818. https://doi.org/10.1111/j.1365-2486.2010.02279.x
- Palmer, S. M., Hope, D., Billett, M. F., Dawson, J. J. C., & Bryant, C. L. (2001). Sources of organic and inorganic carbon in a headwater stream: Evidence from carbon isotope studies. *Biogeochemistry*, 52(3), 321–338. https://doi.org/10.1023/A:1006447706565
- Pawson, R. R., Evans, M. G., & Allott, T. E. H. (2012). Fluvial carbon flux from headwater peatland streams: Significance of particulate carbon flux. Earth Surface Processes and Landforms, 37(11), 1203–1212. https://doi.org/10.1002/esp.3257

Payne, R. J. (2014). The exposure of British peatlands to nitrogen deposition, 1900-2030. Mires & Peat, 14(4), 1-9.

Pronger, J., Schipper, L. A., Hill, R. B., Campbell, D. I., & McLeod, M. (2014). Subsidence rates of drained agricultural peatlands in New Zealand and the relationship with time since drainage. *Journal of Environmental Quality*, 43(4), 1442–1449. https://doi.org/10.2134/jeq2013.12.0505

- Raymond, P. A., McClelland, J. W., Holmes, R. M., Zhulidov, A. V., Mull, K., Peterson, B. J., et al. (2007). Flux and age of dissolved organic carbon exported to the Arctic ocean: A carbon isotopic study of the five largest arctic rivers. *Global Biogeochemical Cycles*, 21(4), GB4011. https://doi.org/10.1029/2007GB002934
- Reimer, P. J., Bard, E., Bayliss, A., Beck, J. W., Blackwell, P. G., Bronk Ramsey, C., et al. (2013). IntCal13 and Marine13 radiocarbon age calibration curves 0–50, 000 years calBP. *Radiocarbon*, 55(4), 1869–1887. https://doi.org/10.2458/azu\_js\_rc.55.16947
- Romanovsky, V. E., Smith, S. L., Christiansen, H. H., Shiklomanov, N. I., Drozdov, D. S., Oberman, N. G., et al. (2012). Permafrost [in 'state of the climate in 2011']. *Bulletin of the American Meteorological Society*, *94*, S1–S238.

Salm, J. O., Maddison, M., Tammik, S., Soosaar, K., Truu, J., & Mander, U. (2012). Emissions of CO<sub>2</sub>, CH<sub>4</sub> and N<sub>2</sub>O from undisturbed, drained and mined peatlands in Estonia. *Hydrobiologia*, 692(1), 41–55. https://doi.org/10.1007/s10750-011-0934-7

- Schuur, E. A. G., & Mack, M. C. (2018). Ecological response to permafrost thaw and consequences for local and global ecosystem services. *Annual Review of Ecology, Evolution and Systematics*, 49(1), 279–301. https://doi.org/10.1146/annurev-ecolsys-121415-032349
- Shurpali, N. J., Hyvönen, N. P., Huttunen, J. T., Biasi, C., Nykänen, H., Pekkarinen, N., & Martikainen, P. J. (2008). Bare soil and reed canary grass ecosystem respiration in peat extraction sites in Eastern Finland. *Tellus B: Chemical and Physical Meteorology*, 60(2), 200–209. https:// doi.org/10.3402/tellusb.v60i2.16915
- Shuttleworth, E. L. (2015). Tracing peatland geomorphology: Sediment and contaminant movements in eroding and restored systems, (Doctoral dissertation). University of Manchester. Retrieved from https://www.research.manchester.ac.uk/portal/en/theses/search.html
- Shuttleworth, E. L., Evans, M. G., Pilkington, M., Spencer, T., Walker, J., Milledge, D., & Allott, T. E. H. (2019). Restoration of blanket peat moorland delays stormflow from hillslopes and reduces peak discharge. *Journal of Hydrology X*, 2, 100006. https://doi.org/10.1016/j. hydroa.2018.100006
- Stimson, A. G. (2015). Fluvial carbon dynamics in degraded peatland catchments (Doctoral dissertation). University of Manchester. Retrieved from https://www.research.manchester.ac.uk/portal/en/theses/search.html
- Stimson, A. G., Allott, T. E. H., Boult, S., & Evans, M. G. (2017). Reservoirs as hotspots of fluvial carbon cycling in peatland catchments. Science of the Total Environment, 580(15), 398–411. https://doi.org/10.1016/j.scitotenv.2016.11.193

- Strack, M., Waddington, J. M., Bourbonniere, R. A., Buckton, E. L., Shaw, K., Whittington, P., & Price, J. S. (2008). Effect of water table drawdown on peatland dissolved organic carbon export and dynamics. *Hydrological Processes*, 22(17), 3373–3385. https://doi.org/10.1002/ hyp.6931
- Tallis, J. H. (1964). Studies on southern Pennine peats: III. The behaviour of Sphagnum. Journal of Ecology, 52(2), 345–353. https://doi.org/10.2307/2257601
- Tallis, J. H. (1985). Mass movement and erosion of a southern Pennine blanket peat. Journal of Ecology, 73(1), 283-315. https://doi.org/10.2307/2259784
- Tallis, J. H. (1987). Fire and flood at Holme Moss Erosion processes in an upland blanket mire. Journal of Ecology, 75(4), 1099–1129. https://doi.org/10.2307/2260317
- Tallis, J. H. (1994). Pool-and-hummock patterning in a Southern Pennine blanket mire II. The formation and erosion of the pool system. Journal of Ecology, 82(4), 789–803. https://doi.org/10.2307/2261444
- Tallis, J. H. (1995). Climate and erosion signals in British blanket peats: The significance of Racomitrium Lanuginosum remains. Journal of Ecology, 83(6), 1021–1030. https://doi.org/10.2307/2261183
- Tallis, J. H. (1997). Peat erosion in the Pennines: The badlands of Britain. Biologist, 44, 277-279.
- Tallis, J. H., & Livett, E. A. (1994). Pool-and-hummock patterning in a Southern Pennine blanket mire I. Stratigraphic profiles for the last 2800 years. *Journal of Ecology*, 82(4), 775–788. https://doi.org/10.2307/2261443
- Tallis, J. H., & Switsur, V. R. (1983). Forest and moorland in the south Pennine uplands in the mid-flandrian period: I. Macrofossil evidence of the former forest cover. *Journal of Ecology*, 71(2), 585–600. https://doi.org/10.2307/2259736
- Tallis, J. H., & Switsur, V. R. (1990). Forest and moorland in the south Pennine uplands in the mid-flandrian period.: II. The hillslope forests. Journal of Ecology, 78(4), 857–883. https://doi.org/10.2307/2260940
- Tipping, E., Billett, M. F., Bryant, C. L., Buckingham, S., & Thacker, S. A. (2010). Sources and ages of dissolved organic matter in peatland streams: Evidence from chemistry mixture modelling and radiocarbon data. *Biogeochemistry*, 100(1–3), 121–137. https://doi.org/10.1007/ s10533-010-9409-6
- Turetsky, M. R., Benscoterm, B., Page, S., Rein, G., van de Werf, G. R., & Watts, A. (2015). Global vulnerability of peatlands to fire and carbon loss. *Nature Geoscience*, 8(1), 11–14. https://doi.org/10.1038/ngeo2325
- van der Werf, G. R., Randerson, J. T., Giglio, L., Collatz, G. J., Mu, M., Kasibhatla, P. S., et al. (2010). Global fire emissions and the contribution of deforestation, savanna, forest, agricultural, and peat fires (1997–2009). Atmospheric Chemistry and Physics, 10(23), 11707–11735. https:// doi.org/10.5194/acp-10-11707-2010
- Worrall, F., Burt, T. P., & Adamson, J. (2006). The rate of and controls upon DOC loss in a peat catchment. Journal of Hydrology, 321(1-4), 311-325. https://doi.org/10.1016/j.jhydrol.2005.08.019